
 
\documentclass[11pt]{report}
 
\usepackage[margin=0.5in]{geometry} 
\usepackage{amsmath,amsthm,amssymb,verbatim}


 
\begin{document}

\large{2015 Spring Math 5043 \quad Assignment 6. \qquad Due: April 24, 2015}
\medskip

%\large{Name: }
\bigskip

Do as many problems as you can from Exercises 2.2 in Hatcher (p. 155).

\smallskip
Hand in the following problems: 4, 9, 10, 11, 14, 15, 19, 29, 30, 31, 32

\smallskip
Note: For 2.2.30, please read Example 2.48 in Hatcher. There will be another assignment after this, but not to be handed in.

\end{document}




















